Author's response to reviews

Title: What Data is Important? Clinical Data Needs in the Neonatal Intensive Care Unit Electronic Medical Record

Authors:

Marc A Ellsworth (ellsworth.marc@mayo.edu)
Tara R Lang (lang.tara@mayo.edu)
Brian W Pickering (pickering.brian@mayo.edu)
Vitaly Herasevich (herasevich.vitaly@mayo.edu)

Version: 6 Date: 19 September 2014

Author's response to reviews:

Dear Editorial Board,

Thank you for the acceptance of our manuscript "What data is important? Clinical data needs in the neonatal intensive care unit electronic medical record," for your journal. We are very excited about the news.

We have added clarification of the ethics approval by our institution for the study. The clarification is contained in the first paragraph of the Methods section (page 4).

Please let us know if any other clarifications or changes need to be made. Thank you again for the acceptance.

Sincerely,

Marc Ellsworth, MD